\begin{thebibliography}{10}

\bibitem{Agarwal91}
A.~Agarwal, D.~Chaiken, G.~D'Souza, K.~Johnson, D.~Kranz, J.~Kubiatowicz,
  K.~Kurihara, B.~H. Lim, G.~Maa, D.~Nussbaum, M.~Parkin, and D.~Yeung.
\newblock {The MIT Alewife Machine: A Large-Scale Distributed-Memory
  Multiprocessor}.
\newblock In {\em Proceedings of Workshop on Scalable Shared Memory
  Multiprocesors}. Kluwer Academic Publishers, 1991.

\bibitem{arvind90}
Arvind and R.~S. Nikhil.
\newblock {Executing a Program on the {MIT} Tagged-Token Dataflow
  Architecture}.
\newblock {\em IEEE Transactions on Computers}, 39(3):300--318, March 1990.

\bibitem{Boughton94}
G.~A. Boughton.
\newblock {Arctic Routing Chip}.
\newblock In {\em Proceedings of Hot Interconnects II, Stanford, CA}, pages 164
  -- 173, August 1994.

\bibitem{Burkhardt92}
H.~{Burkhardt III}, S.~Frank, B.~Knobe, and J.~Rothnie.
\newblock {Overview of the {KSR1} Computer System}.
\newblock Technical Report KSR-TR-9202001, Kendall Square Research, Boston,
  February 1992.

\bibitem{Goodman83}
J.~Goodman.
\newblock {Using Cache Memory to Reduce Processor-Memory Traffic}.
\newblock In {\em Proceedings of The 10th Annual International Symposium on
  Computer Architecture}, pages 124--131, June 1983.

\bibitem{Heinrich94}
M.~Heinrich, J.~Kuskin, D.~Ofelt, J.~Heinlein, J.~Baxter, J.~P. Singh,
  R.~Simoni, K.~Charachorloo, D.~Nakashira, M.~Horowitz, A.~Gupta,
  M.~Rosenblum, and J.~Hennessy.
\newblock {The Performance Impact of Flexibility in the Stanford {FLASH}
  Multiprocessor}.
\newblock In {\em Proceedings of the Sixth International Conference on
  Architecture Support for Programming Languages and Operating Systems, San
  Jose, CA}, pages 274 -- 285, October 1994.

\bibitem{Kranz93}
D.~Kranz, K.~Johnson, A.~Agarwal, J.~Kubiatowicz, and B.-H. Lim.
\newblock {Intergrating Message-Passing and Shared-Memory: Early Experience}.
\newblock In {\em Fourth {ACM} {SIGPLAN} Symposium on Principles and Practice
  of Parallel Programming, San Diego}, pages 54--63, 1993.

\bibitem{Kuskin94}
J.~Kuskin, D.~Ofelt, M.~Heinrich, J.~Heinlein, R.~Simoni, K.~Gharachorloo,
  J.~Chapin, D.~Nakahira, J.~Baxter, M.~Horowitz, A.~Gupta, M.~Rosenblum, and
  J.~Hennessy.
\newblock {The Stanford FLASH Multiprocessor}.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture, Chicago, Il}, pages 302--313, April 1994.

\bibitem{Lenoski90}
D.~Lenoski, J.~Laudon, K.~Charachorloo, A.~Gupta, and J.~Hennessy.
\newblock {The Directory-Based Cache Coherence Protocol for the DASH
  Multiprocesor}.
\newblock In {\em Proceedings of The 17th Annual International Symposium on
  Computer Architecture, Seattle, WA}, pages 148--159, 1990.

\bibitem{Lenoski&others92}
D.~Lenoski, J.~Laudon, T.~Joe, D.~Nakahira, L.~Stevens, A.~Gupta, and
  J.~Hennessy.
\newblock {The DASH Prototype: Implementation and Performance}.
\newblock In {\em Proceedings of The 19th Annual International Symposium on
  Computer Architecture, Gold Coast, Australia}, pages 92--103, May 1992.

\bibitem{Li86}
K.~Li.
\newblock {\em Shared Virtual Memory on Loosely Coupled Multiprocessors}.
\newblock PhD thesis, Yale University, September 86.
\newblock (Also as YALE/DCS/RR-492).

\bibitem{Li89b}
K.~Li and R.~Schaefer.
\newblock {{SHIVA}: An Operating System Transforming a Hypercube Into a
  Shared-memory Machine}.
\newblock CS-TR 217-89, Princeton University, Department of Computer Science,
  April 1989.

\bibitem{Li89a}
K.~Li, M.~Stumm, D.~Wortman, and S.~Zhou.
\newblock {Shared Virtual Memory Accommodating Hetergeneity}.
\newblock CS-TR 210-89, Princeton University, Department of Computer Science,
  February 1989.

\bibitem{Papadopoulos93}
G.~M. Papadopoulos, G.~A. Boughton, R.~Greiner, and M.~J. Beckerle.
\newblock {*T: Integrated Building Blocks for Parallel Computing}.
\newblock In {\em Proceedings of Supercomputing '93, Portland, Oregon}, pages
  624--635, November 1993.

\bibitem{Reinhardt94}
S.~K. Reinhardt, J.~R. Larus, and D.~A. Wood.
\newblock {Tempest and Typhoon: User-Level Shared Memory}.
\newblock In {\em Proceedings of the 21st Annual International Symposium on
  Computer Architecture, Chicago, Il}, pages 325--336, April 1994.

\bibitem{shen94}
X.~W. Shen.
\newblock {Global Cache Coherence Schemes for \start/}.
\newblock {CSG} Memo 361, {Laboratory for Computer Science, MIT}, Cambridge MA,
  December 1994.

\bibitem{Thacker87}
C.~P. Thacker, L.~Stewart, and J.~E.H.~Satterthwaite.
\newblock {Firefly: a multiprocessor workstation}.
\newblock TR~23, DEC/SRC, Digital Equipment Corporation, Systems Research
  Center, December 1987.

\end{thebibliography}

%%------------------------------------------------------------------------
\section{Introduction}
%%------------------------------------------------------------------------
\label{sec:intro}
%%------------------------------------------------------------------------

This paper describes the challenges we have faced in designing a
parallel machine based on commercial SMPs.  We, like most other
researchers, believe that to support a general programming model, a
machine must provide both a global address space and efficient message
passing for short messages.  From an implementation perspective, there
is some synergy between these two requirements because creating the
illusion of a global address space across a network of SMPs or
workstations requires efficient message passing for cache-line size
messages.  However, due to the presence of caches in microprocessors,
supporting shared memory abstraction on a distributed memory machine
is significantly harder than pure message passing.

Researchers have devised a whole spectrum of techniques to implement a
shared address space on distributed memory machines.  At one extreme,
Li\cite{Li86,Li89a,Li89b} has implemented coherent shared memory at
page-level granularity without any hardware support beyond message
passing.  He does so by modifying the virtual memory manager to allow
demand paging from remote nodes.  This scheme has been reported to
work well in certain application areas.

At the other extreme are machines that provide extensive hardware
support for shared memory.  Some machines, such as the Cray T3D, allow
the user to access global memory in a cached or uncached fashion;
however, the hardware does not maintain cache coherence.  Machines
that support cache coherence beyond single-bus systems invariably
provide some processing power to execute directory-based coherence
protocols.  Machines in this class include the Stanford
\dash/\cite{Lenoski90,Lenoski&others92}, the Kendall Square Research
KSR1\cite{Burkhardt92}, and the MIT \alewife/\cite{Agarwal91}, which
are all operational, and machines under design such as the Stanford
\flash/\cite{Kuskin94} and the University of Wisconsin \typhoon/(a
paper architecture).  

Our project, called \start/, is an attempt to build an efficient
message passing machine using commercial SMPs based on PowerPC 620's.
Each processor directly connects to an MPP-class network via a network
interface unit.  The memory bus in each SMP is augmented with a simple
address capture device (ACD) which can be used in conjunction with one
of the processors to support various shared memory models.  Each SMP
{\em site} modified in this manner will still run a standard operating
system (OS) such as an SMP version of Unix.  We have chosen to run an
OS image per site as opposed to a single image on the whole system for
better fault tolerance.

In addition to supporting all kinds of message-passing applications,
we would like to run parallel applications written for SMPs on
\start/.  This requires that we implement cache coherent global memory
across sites.  We think that \start/ will provide an inexpensive,
low-risk, and flexible experimental vehicle for exploring coherence
protocols for various memory models including I-structures\cite{},
distributed operating system issues, and application development in a
hybrid message passing/coherent shared memory environment.

The main contribution of this paper is the identification of some
specific problems in implementing cache-coherent distributed shared
memory (\ccdsm/) on commercial SMPs.  The problems we will discuss
have origins in three different sources: the processor and its snoopy
bus protocol, buffering within the system and the non-FIFO character
of the network.  Though the problems are discussed in the context of
PowerPC, we think some of these problems are likely to be present in
any parallel coherent shared memory system based on stock
microprocessors.  As we will show, there are software solutions around
most of these problems.  However, with a slightly different processor
or bus design, some of these problems can either be completely
eliminated or their solutions made much simpler.  We hope that these
findings may influence the design of future processors and snoopy bus
protocols.

In the next section, we will give the background information necessary
to understand the challenges we faced.  It will describe particular
aspects of processors, memory buses, buffering and the network without
getting into the problems they may cause.  Section~\ref{sec:start}
describes the \start/ implementation and how we plan to implement
\ccdsm/ with a small piece of custom hardware and a dedicated
processor.  Estimates for the performance of \start/'s \ccdsm/ are
included in that section.  In Section~\ref{sec:ccdsm-issues}, we
discuss the challenges that we encountered while implementing \ccdsm/.
The final section gives some conclusions.

